Author's response to reviews

Title: Non-specific physical symptoms in relation to actual distance and perceived proximity to mobile phone base stations and powerlines.

Authors:

Christos Baliatsas (Christos.Baliatsas@rivm.nl)
Irene Van Kamp (irene.van.kamp@rivm.nl)
Gert Kelfkens (Gert.Kelfkens@rivm.nl)
Maarten Schipper (maarten.schipper@rivm.nl)
John Bolte (John.Bolte@rivm.nl)
Joris Yzermans (J.Yzermans@nivel.nl)
Erik Lebret (erik.lebret@rivm.nl)

Version: 4 Date: 31 May 2011

Author's response to reviews: see over
Author’s response

Title: “Non-specific physical symptoms in relation to actual and perceived proximity to mobile phone base stations and power lines.”

Authors:

Christos Baliatsas (Christos.Baliatsas@rivm.nl)
Irene Van Kamp (irene.van.kamp@rivm.nl)
Gert Kelfkens: (Gert.Kelfkens@rivm.nl)
Maarten Schipper (maarten.schipper@rivm.nl)
John Bolte (John.Bolte@rivm.nl)
Joris Yzermans (J.Yzermans@nivel.nl)
Erik Lebret (erik.lebret@rivm.nl)

Version: 2 Date: 31 May 2011

Author’s response: please see below (in blue colour)
Final formatting change suggested by the editorial office:

Tables: Please ensure that the tables are cited in the manuscript in numerical order. Table 2 is currently cited first.

The tables are now cited in numerical order.